Author's response to reviews

Title: Choosing algorithms for TB screening: a modelling study to compare yield, predictive value and diagnostic burden.

Authors:

Anna H van't Hoog (a.h.vanthoog@aighd.org)
Ikushi Onozaki (onozakii@who.int)
Knut Lonnroth (lonnrothk@who.int)

Version: 2 Date: 7 April 2014

Author's response to reviews:

To: the editorial office

Dear madam/sir,

Following your request for changes, the following revisions have been made:

1. Line Numbering has been added.

2. An ethics statement has been added, explaining that “Ethical approval was not sought, as only secondary data were used.”
   This is because it is a modeling study, and we took the parameter values from the literature.

3. Acknowledgements: this section has been added and includes acknowledgement of the funding organization and it’s role.

I hope that this sufficiently addresses the points raised.

Sincerely,

Anna H. van't Hoog